Author's response to reviews

Title: Detecting imipenem resistance in Acinetobacter baumannii by automated systems (BD Phoenix, Microscan WalkAway, Vitek 2); high error rates with Microscan WalkAway

Authors:

Canan Kulah (canankulah@yahoo.com)
Elif Aktas (drelifaktas@yahoo.com)
Fusun Comert (fusbeg@yahoo.com)
Nagihan Ozlu (drozlu2003@yahoo.com)
Isin Akyar (iakyar@acibademlabmed.com.tr)
Handan Ankarali (hankarali@yahoo.com)

Version: 4 Date: 5 March 2009

Author's response to reviews: see over
Dear Editor,

We are glad that our paper entitled “Detecting imipenem resistance in Acinetobacter baumannii by automated systems (BD Phoenix, Microscan WalkAway, Vitek 2); high error rates with Microscan WalkAway” was considered for publication.

We finally checked the manuscript for typing errors.

We also checked that our manuscript conforms to the journal style.

Formatting changes are done as requested as follows;

References - The reference list is checked and corrected.

Reference 11 full details for this reference is added.

Reference 12 full details for this reference is added.

Reference 13 full details for this reference is added.

Table shading - Shading is excluded.

Additional files - A section titled "Additional files" is added at the end of the manuscript (after the tables). The authors declare that there are no additional files.

Sentence case for the authors' names are used

Keywords are removed.

Yours sincerely,

Dr. Canan Kulah, MD